\documentstyle[12pt]{article}

\begin{document}

\section*{Topics for Week of November 8th, 1999}

\subsection*{Combinations\\
Rosen 4.3, Tutorial Notes 10/24, Fall 97}
\begin{itemize}
\item
Counting Combinations
\item
Binomial Coefficents
\item
Pascal's Triangle
\item
The Binomial Theorem
\end{itemize}

\subsection*{$r$-Combinations\\
F97 Lec Notes 15 }

\begin{itemize}
\item
The division Rule 
\item
Examples of $r$-Combinations
\end{itemize}

\subsection*{Counting with Repetitions \\
F97 Lec Notes 16, Rosen 4.6 }
\begin{itemize}
\item
Counting Combinations and $r$-combinations with Repetitions
\item 
Examples of the Use of Combination with Repetitins:
data compression, stars and balls
\item
$r$-combinations with at least one item of each kind
\item
$r$-combinations with limited Repetitions
\item
Counting Permutations and $r$-Permutations  with Repetitions
\item
Counting Permutations with Limited Repetitins 
\item
Multinomial Coeficients 
\item
Permutations of Sets with Indistinguishable Objects
\end{itemize}


\end{document}




